Author's response to reviews

Title: Semiallogenic fusions of MSI+ tumor cells and activated B cells induce MSI-specific T cell responses

Authors:

Yvette Garbe (yvette.Garbe2@uniklinikum-dresden.de)
Ulrike Klier (ulrike.klier@med.uni-rostock.de)
Michael Linnebacher (michael.linnebacher@med.uni-rostock.de)

Version: 4 Date: 13 September 2011

Author's response to reviews:

Dear Sirs and Madams,

I have currently no access to my full data store. This is why I write a cover letter for the proofs of our manuscript into the online-system only.

However, I made all the changes requested from us:

- the references are now cited in the numerical order correctly.
- the tables are now sized for a standard page. Specifically, Table 2 was adapted accordingly.
- all additional files have been removed from the system. They have been included for the revision process and are not needed in the final version of the manuscript.

I also made the minor revisions you have asked:

- all trackings were removed from the final version of the manuscript.
- the image files have been cropped closely. Additionally, all figure numbers have been removed from the image files.

Of course I carefully checked the manuscript for typographical errors. These have not been highlighted.

Thank you for accepting our manuscript for publication in BMC Cancer and all the best

Michael Linnebacher